Author's response to reviews

Title: The development and implementation of a regional network of physiotherapists for exercise therapy in patients with peripheral arterial disease, a preliminary report

Authors:

Edith M Willigendael (willigendael@hotmail.com)
Bianca LW Bendermacher (biancabenermacher@hotmail.com)
Eduard van de Berg (c.vanderberg@atriummc.nl)
Rob RJTHJ Welten (jwelten@knmg.nl)
Martin H Prins (mh.prins@epid.unimaas.nl)
Rob A de Bie (RA.deBie@epid.unimaas.nl)
Joep AW Teijink (j.teijink@atriummc.nl)

Version: 5 Date: 9 March 2005

Author's response to reviews: see over
Dear Editor,

Thank you for accepting our manuscript to the BioMed Central Health Services Research, with the title:

‘The development and implementation of a regional network of physiotherapists for exercise therapy in patients with peripheral arterial disease, a preliminary report’

The suggested minor second revisions by the reviewer have been implemented.

We have changed: ‘the quality of care’ into the ‘access to care’.

The second suggestion was to include the patient’s leaflet on exercise therapy is unfortunately not possible, since it is a booklet of 28 pages, we assume this is beyond the scope of the journal.

We have implemented the editorial changes, (like removing the table numbers and legends from the figure files), but we have changed the normal design of ‘background, methods and results’ into different sections as has been suggested by the previous report of the reviewer. We hope that this will not be a problem.

All authors have seen and approved the final version of the manuscript and stated that no financial or other conflict of interest are present. The authors have fulfilled the criteria of authorship and assign hereby copyright to the BioMed Central.

On behalf of all authors,
Yours sincerely,

Joep AW Teijink, MD, PhD
Atrium Medical Centre
Department of Surgery, Division of Vascular Surgery
PO Box 4446
6401 CX Heerlen
The Netherlands
Phone: 0031 (0) 45 5766599
Fax: 0031 (0) 45 5766548
E-mail: j.teijink@atriummc.nl